

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Arbel, Yuval; Fialkoff, Chaim; Kerner, Amichai

Article

Can the Laffer curve become a policy tool for reducing tax evasion? Stratification of property tax collection by land use

Journal of Urban Management

Provided in Cooperation with:

Chinese Association of Urban Management (CAUM), Taipei

Suggested Citation: Arbel, Yuval; Fialkoff, Chaim; Kerner, Amichai (2019) : Can the Laffer curve become a policy tool for reducing tax evasion? Stratification of property tax collection by land use, Journal of Urban Management, ISSN 2226-5856, Elsevier, Amsterdam, Vol. 8, Iss. 1, pp. 57-74, https://doi.org/10.1016/j.jum.2018.09.001

This Version is available at: https://hdl.handle.net/10419/271337

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



ND https://creativecommons.org/licenses/by-nc-nd/4.0/





Contents lists available at ScienceDirect

Journal of Urban Management

journal homepage: www.elsevier.com/locate/jum



Can the Laffer curve become a policy tool for reducing tax evasion? Stratification of property tax collection by land use^{\star}



rallu 🐨

Yuval Arbel^{a,*}, Chaim Fialkoff^b, Amichai Kerner^c

^a Sir Harry Solomon School of Management, Western Galilee College, Acra 2412101, Israel

^b Institute of Urban and Regional Studies, Hebrew University of Jerusalem, Mt. Scopus, Jerusalem 91905, Israel

^c School of Real Estate, Netanya Academic College, 1 University Street, Netanya 4223587, Israel

ARTICLE INFO

JEL codes: H71 R51 R52 Keywords: Local municipal tax Laffer curve Land uses

ABSTRACT

We construct Laffer curves to evaluate the efficiency of local property tax collection based on a micro-level panel dataset referring to 2013–2016 and obtained from the Jerusalem municipality. Unlike previous literature, we apply a tractable ad hoc methodology, which, compared with the conventional approach, might prove to be more useful to practitioners and policy makers. Results indicate the efficiency of a tax policy designed to increase tax rates for on-sample taxpayers. According to one estimate, to preserve the same collected sum of payments as before, for the entire sample, the minimal anticipated required raise is 4.17% per annum.

1. Introduction

The Laffer curve is a hump-shaped curve describing the relationship between accumulated tax revenues and tax rates. While the idea that people respond to increasing tax rates by tax avoidance (e.g., diminishing consumption) or tax evasion (e.g., encouraging smuggling) is quite familiar to economists,¹ the more controversial claim of Laffer is that the current tax rate is greater than the optimal tax rate, which maximizes revenues (Fullerton, The New Palgrave Dictionary of Economics, 2008).²

* Corresponding author.

https://doi.org/10.1016/j.jum.2018.09.001

Received 12 June 2018; Accepted 7 September 2018

Available online 02 October 2018

Peer review under responsibility of Zhejiang University and Chinese Association of Urban Management.

^{*} The authors would like to thank the Jerusalem Municipality and Anat Chechik for provision of project data and other materials, and Yifat Arbel, and the participants of the 2017 Regional Science Association Meeting for helpful comments.

E-mail addresses: YuvalAr@wgalil.ac.il (Y. Arbel), chaim.fialkoff@mail.huji.ac.il, cfialk@gmail.com (C. Fialkoff), kerneram@netvision.net.il (A. Kerner).

¹ A nice example of tax avoidance is given in Fischel, 1992. Given that in the rural town of Orford, New Hampshire, constructing a barn increases property tax charge, the local farmers used Scottish Highland cattle breed, which do not have to be kept in a barn during winter. The author states that: "The deadweight loss of the property tax is hairy cattle." (page 171). In addition, there is a vast empirical and experimental literature on tax evasion and tax auditing. For reviews of the literature see Andreoni, Erard and Feinstein, (1998); and Kirchler, 2007. Two subsequent examples are: Kastlunger et al., 2009; Rees-Jones, 2018.

² The origin of the term "Laffer curve" is Wanniski, 1978. As Laffer, 2004 notes: "As recounted by Wanniski (associate editor of The Wall Street Journal at the time), in December 1974, he had dinner with me (then professor at the University of Chicago), Donald Rumsfeld (Chief of Staff to President Gerald Ford), and Dick Cheney (Rumsfeld's deputy and my former classmate at Yale) at the Two Continents Restaurant at the Washington Hotel in Washington, D.C. While discussing President Ford's "WIN" (Whip Inflation Now) proposal for tax increases, I supposedly grabbed my napkin and a pen and sketched a curve on the napkin illustrating the trade-off between tax rates and tax revenues. Wanniski named the trade-off "The Laffer Curve."

^{2226-5856/ © 2019} Zhejiang University and Chinese Association of Urban Management. Production and hosting by Elsevier B.V. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/BY-NC-ND/4.0/).

Table 1	
Distribution and Anticipated and Actual Local Tax Collection 2013–2016 Stratified by Land Use.	

VARIABLES	Frequency	%	Actual Payment (\$)	%	Total Net Tax Charge (\$)	%
OWNER_OCCUPIER (RESIDENCE)	54,064	61.41%	160,842,636.35	61.18%	215,427,626.60	60.45%
RENTERS (RESIDENCE)	29,963	34.03%	45,453,285.80	17.29%	72,575,524.50	20.36%
OFFICES_BELOW_150	2838	3.22%	20,813,306.12	7.92%	24,279,588.04	6.81%
OFFICES_ABOVE_150	410	0.47%	20,662,129.95	7.86%	24,154,888.80	6.78%
WORKSHOP_GARAGES	317	0.36%	3,292,817.00	1.25%	3,746,709.75	1.05%
INDUSTRY	123	0.14%	3,156,841.35	1.20%	3,811,334.55	1.07%
HOTELS	26	0.03%	3,036,127.36	1.15%	4,165,223.68	1.17%
INSTITUTIONS	202	0.23%	2,157,560.72	0.82%	3,642,860.16	1.02%
PARKING_LOTS	11	0.01%	1,645,250.03	0.63%	2,319,727.10	0.65%
BOWLING_GYMS	21	0.02%	575,923.68	0.22%	754,467.36	0.21%
GOVERNMENT	7	0.01%	485,923.69	0.18%	498,987.58	0.14%
CINEMA_THEATERS	9	0.01%	228,579.79	0.09%	289,798.49	0.08%
INSURANCE	8	0.01%	181,924.60	0.07%	191,455.20	0.05%
OCCUPIED_LAND	21	0.02%	129,974.88	0.05%	175,503.86	0.05%
BANQUETTE_HALLS	2	0.00%	117,694.92	0.04%	198,528.98	0.06%
BANKS	3	0.00%	78,492.70	0.03%	81,341.45	0.02%
DANCE_CLUBS	1	0.00%	19,858.58	0.01%	20,145.13	0.01%
ATAROT_WAREHOUSES	1	0.00%	10,500.00	0.00%	27,833.88	0.01%
ARTIST_STUDIOS	10	0.01%	8875.00	0.00%	14,936.73	0.00%
FREE_PARKING	1	0.00%	387.01	0.00%	390.92	0.00%
Observations	88,033	100.00%	262,898,089.53	100.00%	356,376,872.76	100%

Notes: The budget is originally collected in NIS (the local Israeli currency). All the figures are converted to US Dollars, where 1 NIS roughly equals \$0.25. The sum in US Dollars for each land use is calculated as the product between the number of observations and the sample mean. Land uses are sorted in descending order based on the respective share of revenues from each of them.

Empirical researchers have attempted to estimate this optimal income tax rate by developing two prominent empirical methodologies. The earlier methodology was based on simple models of leisure and home production (Stuart, 1981) and estimates or assumptions made concerning labor supply and demand elasticities (e.g., Fullerton, 1982). The latter approach used tax reforms as natural experiments by observing the response of individuals to these tax reforms. A common measure employed in these studies, *e*, is defined as the elasticity of taxable income with respect to a change in the marginal net of tax share ((1 - t), where t is the marginal $tax rate). If the economy had only a single tax rate, the revenue maximizing tax rate is simply <math>t^* = \frac{1}{1+e}$ (Fullerton, The New Palgrave Dictionary of Economics, 2008).

Feldstein, 1995 examines the fall of the highest marginal tax bracket (from 50 percent to 28 percent), and high and middle marginal tax brackets following the 1986 Tax reform Act in United States based on panel data of 3792 individuals (see Table 1, page 561). He calculated *e* for each marginal tax bracket, and found that e = 1.04 - 1.45, if we offset partnership tax losses (see Table 2, page 565). This implies $t^* = 40.82\% - 49.0\%$. Gruber and Saez (2002) used a 1979–1990 panel of tax returns to analyze *all* state and Federal tax reforms during the 80 s, and found that e = 0.4, which implies $t^* = 71.43\%$. This estimate is similar to those obtained by the earlier methodology, and seems to be consistent with Fullerton's view for an over-simplified world (Fullerton, 2008).³

Another strand of the literature is the attempt to estimate the VAT Laffer curve. Oliviera, Francisca and Costa (2015) examine a panel data of 27 European states in 1995 and 2000–2011. By using regression analysis, and a time variable (equals 1 in 1995, and 6–12 for 2000–2011), the authors distinguished between periods of economic recession and expansion, and demonstrated that the optimal VAT decreased from 25.0% in expansion periods (1995) to 22.0% in expansion periods (2011). The equivalent figures in recession periods are 23.50% (1995) and 21.50% (2011).

Very few studies examined the efficiency of tax collection at the municipal level. Based on a time series of aggregated data, Haughwout, Inman, Craig, Steven and Luce (2004) constructed Laffer curves for four U.S. cities (Houston, Minneapolis, New York and Philadelphia). Yet, unlike micro-level panel data, the use of aggregated data is associated with information-loss and might be subject to aggregation bias.⁴ Moreover, this approach might prove to be less useful to practitioners and policy makers in the municipal level who, rather than tax charge, are more interested in actual tax collection.

We propose and apply a different methodology to estimate the Laffer curve. The two key variables in our analysis are the actual tax collection, which is the main interest of local municipalities, compared with the actual tax charge. We thus construct Laffer Curves directly by estimating the elasticity of annual payment with respect to the annual local property tax charge for different land uses and across time. The parabolic shape of the curve permits this elasticity to vary with the level of tax charge.

We employ a unique panel dataset referring to 2013–2016 obtained from Jerusalem municipality. The dataset contains microlevel information of the land use, the annual local property tax charge and payment of each individual taxpayer, and the area of each asset measured in square meters. To avoid extraordinary cases of coverage of debt in arrears, we restrict the sample only to taxpayers,

³ For a survey of a more recent literature, see, for example, Saez, Slemrod and Giertz (2012).

⁴ For a discussion on the different estimates obtained in aggregated and micro level housing demand data see, for example, Mayo, 1981.

Table 2

Average Collection Rates for Different Land Uses 2013-2016.

VARIABLES	Frequency	%	$AVG\left[\frac{PAYMENT}{ARNONA} ight]$
OWNER_OCCUPIER(RESIDENCE) RENTERS(RESIDENCE) OFFICES_BELOW_150 OFFICES_ABOVE_150 WORKSHOP_GARAGES INDUSTRY HOTELS INSTITUTIONS PARKING LOTS	54,064 29,963 2838 410 317 123 26 202 11	61.41% 34.03% 3.22% 0.47% 0.36% 0.14% 0.03% 0.23% 0.01%	94.13% 85.25% 86.52% 86.52% 86.94% 90.61% 82.80% 76.80% 85.02% 72.32%
BOWLING GYMS GOVERNMENT CINEMA_THEATERS INSURANCE OCCUPIED_LAND BANQUETTE_HALLS BANKS DANCE_CLUBS ATAROT_WAREHOUSES ARTIST_STUDIOS FREE_PARKING Observations	21 7 9 8 21 2 3 1 1 1 10 1 88,033	0.02% 0.01% 0.01% 0.02% 0.00% 0.00% 0.00% 0.00% 0.01% 0.00%	80.31% 96.14% 85.83% 86.35% 76.21% 60.08% 95.29% 98.58% 37.72% 75.23% 99.00% 91.22%

Notes: The $\left[\frac{PAYMENT}{ARNONA}\right]$ is the collection rate namely payment-property-tax-charge ratio, which is calculated to each observation separately. The $AVG\left[\frac{PAYMENT}{ARNONA}\right]$ is the average across all observations with the same land use.

who paid between 0% to 100% of the annual property tax charge. The local property tax can be examined in three dimensions: temporally, cross sectional based on different land uses, and geographically (West vs. East Jerusalem). During 2013-2016, the Jerusalem municipality steadily increased the property tax rate by 0.46%-1.42% per annum. Our study thus permits the examination of the response of taxpayers to those increases, and provides efficiency evaluation of this policy.

The main contribution of the current study lies in the application of a tractable and direct methodology that may become a useful tool for municipal policy makers. This methodology enables planning for the future tax policy, as well as improving the efficiency of local property tax collection based on actual collection tax data. The proposed methodology requires only a limited number of variables, including annual tax charge, annual tax collection, area and land use of the asset. Unlike empirical studies at the macro level, the proposed model is not limited to a single tax rate, and can provide different predictions to each land use and asset based on empirical data.

Results indicate that on the one hand, for most taxpayers in land uses such as residence, office, hotel and parking lot, tax collection is expected to rise significantly following a 10% increase in local property tax. On the other hand, within the on-sample range, the elasticity of payment with respect to local property tax charge is quite low. For the entire sample, a 10% increase in afterdiscount (before-discount) tax charge is expected to yield only 3.625–5.768% (1.538- 3.064%.) increase in tax collection.⁵ Nevertheless, these figures imply that raising the tax rate is efficient. For a \$90 payment of a current after-discount net tax charge of \$100, a 10% rise to \$110 is associated with a minimum and maximum anticipated rise to \$93.26 and \$95.19, respectively.⁶ Despite the drop in anticipated collection rates from 90% to 84.78-86.54%, the collected sum is nevertheless expected to rise.⁷

Moreover, additional outcomes obtained from direct estimation of payment-property tax charge elasticities show that on the one hand, if local property tax charge remains unchanged, collection rates are anticipated to drop by 3.37% per annum. On the other hand, if the municipality prefers to preserve the same collected sum of payments as before, for the entire sample, the minimal anticipated required raise is 4.17% per annum. Finally, within the on-sample range (between -5% and 10% variation in the net local property tax), the elasticity of payment-property tax charge is around 0.8. This implies that the projected increase in payment following a 10% increase in after-discount net local property tax charge is approximately 8%.

Referring to the geographical dimension of West vs. East Jerusalem, following a 10% increase of the net after-discount annual

⁵ Discounts are given on the basis of economic and socio-demographic criteria. Given that Jerusalem is one of the poorest cities in Israel, many taxpayers, particularly in the residential sector, receive these discounts. From the taxpayer's perspective, the relevant tax charge is the net charge after discounts.

⁶ As demonstrated in Table 2, the initial tax collection rate for all sectors is 91.22% ⁷ The minimum rise is: $\$90 \cdot \left[1 + \frac{3.625}{100}\right] = \93.2624 and the maximum rise is $\$90 \cdot \left[1 + \frac{5.768}{100}\right] = \95.1913 . The minimum and maximum collection rates are: $\frac{93.2624}{110} = 84.78\%$ and $\frac{95.1913}{110} = 86.54\%$, respectively. All the calculations are based on the assumption that the area of the asset remained un-changed during the sample period.

local property tax tariffs of \$10, \$15, \$20 and \$25 per square meter, payments of taxpayers in West (East) Jerusalem are expected to rise by 3.554%, 4.498%, 5.18% and 5.695% (5.761%, 6.48%, 6.847%, 7.027%). Consequently, compared with West Jerusalem, for a given tax tariff, raising the tax charge in East Jerusalem produces higher revenues.

Referring specifically to the residential land use, according to our calculations, this land use is expected to yield the highest revenues for the municipality (61.18% of the total anticipated collection for owner-occupiers and 17.29% of the total anticipated collection for renters). Results support those obtained in Arbel, Fialkoff and Kerner (2017), according to which compared with renters, the level of tax compliance is higher among owner-occupiers. Following a 10% increase of the after-discount net annual local property tax tariffs per square meter of \$18, \$20, \$22 and \$24, payments of owner-occupiers are expected to rise by 7.078%, 7.064%, 7.023% and 6.961%, and payments of renters are expected to increase by 6.504%, 6.491%, 6.446% and 6.375%.

Referring specifically to land use of offices below and above 150 m^2 , according to our calculations, these land uses are expected to yield the second highest revenues after residential use for the municipality (7.92% and 7.86% of the total anticipated collection for offices below and above 150 m^2 , respectively). Also, compared with residential land use, the after-discount net property tax tariffs of office land uses are much higher (\$81-\$84 compared with only \$18-\$24 per square meter). Results indicate that the efficiency associated with a rise in local property tax tariffs in the office sector is much higher among big offices. Following a 10% increase of the after-discount net annual local property tax tariffs of \$83, \$84, \$85 and \$86 per square meter, payments of big offices with area above 150 m² are expected to rise by 14.756%, 14.848%, 14.938% and 15.026%, ⁸ and payments of small offices with area below 150 m² are expected to increase by only 1.960%, 1.965% and 1.966%.⁹

The rest of this paper is organized as follows. Section 2 provides background for the municipal tax system in Jerusalem, Israel. Section 3 discusses the Laffer Curve in an aggregated level of all land uses. Section 4 stratifies the micro-level dataset based on different land uses, and constructs Laffer curves for selected land uses. Section 5 provides robustness tests by estimating directly the payment-local tax charge elasticity, while considering the actual increases and reductions in tax rate tariffs in 2013–2016. Finally, Section 6 concludes and summarizes.

2. Background: local property tax in Jerusalem

Jerusalem is Israel's capital and the country's largest city with a population of almost 865,000 persons at the end of 2015.¹⁰ According to the Israeli Central Bureau of Statistics (CBS), stratification of population based on religion shows that the two major populations are Jews (61.4%) and Arab-Muslims (35.6%). These populations are economically segregated and also live in predominantly separate geographical locations, namely West and East Jerusalem (Alperovich & Deutsch, 1996). Until the Six Day War in 1967, East Jerusalem was part of Jordanian Kingdom. Since its annexation, the inhabitants of East Jerusalem were granted full municipal rights. Still, and as shown below, East Jerusalem suffers from relatively low municipal investment and relatively higher poverty rates.

Jerusalem is also one of the poorest cities in Israel. According to the Israeli CBS, on a scale of 0–10 with 10 being the highest rating, the socio-economic cluster of Jerusalem is only 3. The Israeli CBS performs a socio-economic-ranking of the 255 cities and towns in Israel based on factor analysis using a large number of variables. Among other objectives, one of the purposes for generating such an index is the provision of a criterion for the allocation of resources to these authorities. Yitzhaki et al. (2013) compare between the current ranking and that which would be obtained if the ranking based solely on the income variable. Given the availability of this variable and the anticipated high correlation with other socio-demographic variables, the comparison is important. The main finding of their study is that there is little difference in the ranking of municipal authorities between that based solely on mean per-capita income and that based on additional variables.

The two ranking criteria are the socio-economic cluster of the population (on a scale from 1 - the lowest, to 10 - the highest); and the relative socio-demographic location of each city (from 1 - the poorest to 255 - the wealthiest). The map in Appendix B describes the clusters of all geographical units in Israel.¹¹ Jerusalem is ranked in cluster 3 (a drop of one place from previous years); and the socio-demographic location is 61, which is below 194 of the 255 ranked cities, namely, the lowest quartile in terms of wealth.¹²

The high poverty rate in Jerusalem is reflected in the high percentage of households who received partial discounts or total waivers of their local residential property tax obligation for residential use. Appendix C presents the 2015 percentage of discount on local property tax in Jerusalem stratified by income levels. Discounts are determined based on the number of persons living in the dwelling unit, and the average monthly income during the last quarter of 2014. Descriptive statistics available upon request show

⁸ Based on the collection rates from the office sector, for a \$85 payment of a current after-discount net tax charge of \$100, a 10% rise to \$110 is associated with a minimum and maximum anticipated rise to $\$5 \cdot \left[1 + \frac{14.756}{100}\right] = \97.5426 and $\$5 \cdot \left[1 + \frac{15.026}{100}\right] = \97.7721 , respectively. These figures imply an increase in collection rates from 85% to $\frac{97.5426}{110} = \$8.675\% - \frac{97.7721}{97.721} = \8.8837%

⁹ Based on the collection rates from the office sector, for a \$85 payment of a current after-discount net tax charge of \$100, a 10% rise to \$110 is associated with a minimum and maximum anticipated rise to $\$85 \cdot \left[1 + \frac{1.960}{100}\right] = \86.666 and $\$85 \cdot \left[1 + \frac{1.966}{100}\right] = \86.6711 , respectively. These figures imply a drop in collection rates from 85% to $\frac{\$6.666}{110} = 78.787\% - \frac{\$6.6711}{110} = 78.792\%$

¹⁰ The map in Appendix A shows that Jerusalem is located in region 1.

¹¹ The map is available at: http://www.cbs.gov.il/hodaot2013n/24_13_087map.pdf

¹² We calculated the quartile as: $\frac{194}{255} = 76.08\%$

that of the total 83,751 residential taxpayers in the sample, 43,627 households (a majority of 52.09%) received a discount, which is equal to or above 4% (the minimum discount rate for the non-residential sector). The median and average discount rate for this group of 43,627 households is 46.56% and 50.11%, respectively.

Appendix D presents the annual local property residential tax tariffs per square meter in NIS in 2013 and 2017.¹³ The three criteria to determine these tariffs are the building types, the geographic zone (A, B, C or D), and the unit's area in square meters. The appendix implies a 7.25%-7.31% increase in local tax charge per square meter during a period of five-years. This implies a $[(1 + 0.0725)^{0.2} - 1] = 1.41\%^{-}$ a $[(1 + 0.0731)^{0.2} - 1] = 1.42\%$ increase per annum.

Finally, Appendix E displays the annual local non-residential property tax tariffs per square, meter in NIS in 2013 and 2017. With the exception of offices, with a relatively high charge per square meter, the table indicates a 7.25%-7.31% increase, which is quite similar to the increase obtained in the residential sector. Unlike the residential sector, however, of the 3984 panels belonging to other sectors, the vast majority of 3885 panels (consisting of 97.52%) receive no discounts at all, and only 99 panels (consisting of 2.48%) receive discounts. The average discount given to these few panels is 57.89% and the minimum and maximum discounts are 4.11% and 67% respectively.

3. An aggregated Laffer curve

To estimate the coefficients of the Laffer curve, we propose and apply the following fixed-effect model:

$$Y_{i,t} = A_1 X_{i,t}^2 + B_1 X_{i,t} + C_1 + \mu_{1,i,t}$$
(1)
$$\mu_{1,i,t} = \alpha_{1,i} + \epsilon_{1,i,t}$$
(2)

Where *i* is the index for each panel (individual or asset); *t* is the time index (t = 2013, 2014, 2015, 2016); $Y_{i,t} = (PAYMENT_{i,t})$, the annual payment; $X_{i,t} = \begin{bmatrix} \frac{ARNONA_{i,t}}{AREA_{i,t}} \end{bmatrix}$, where $ARNONA_{i,t}$ is the net annual local property tax charge for asset *i* at year *t* converted to US dollars (exchange rate of 1 NIS = \$0.25), ¹⁴ and $AREA_{i,t}$ is the area of asset *i* at year *t* measured in square meters (1 m² = 10.7639) square feet), A₁, B₁, C₁ are parameters, $\alpha_{1,i}$ reflects a row vector of parameters (one parameter for each panel), and $\mu_{1,i,t}$ and $\epsilon_{1,i,t}$ are the random disturbance terms. The parabolic model defined by Eq. (1) enables us to calculate the maximal point $\left(-\frac{B_1}{2a_1}\right)$, and the

elasticity of $Y_{i,t}$ with respect to $X_{i,t} \left(E_{i,t} = \frac{2A_1 X_{i,t}^2 + B_1 X_{i,t}}{A_1 X_{i,t}^2 + B_1 X_{i,t} + C_1} \right)$ Referring to Eq. (2), if the fixed-effect model holds, each individual or asset *i* has a different $\alpha_{1,i}$. If these differences across panels exist, but are ignored, then the estimation results may be biased and inconsistent (see, for example, Johnston and Dinardo, 1997: 395–396; Greene, 2012: 399–402).¹⁵ Consequently, we ran two statistical tests to support the fixed-effect model: 1) F-test for the null hypothesis of equal coefficients across all panels. 2) The Wu-Hausman procedure, which tests the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (e.g., Johnston and Dinardo, 1997: 403-404. Greene, 2012: 419-421).

Fig. 1 displays the 2013–2016 aggregated Laffer curve of all land uses at the top of the figure, and the horizontal box-plot after omission of outliers at the bottom of the figure. The horizontal axis of the Laffer curve and the box-plot is the ARNONA (annual net local property tax charge) divided by the area, which is measured in square meters. The vertical axis of the Laffer curve is the projected 2013–2016 annual payment. For the sake of simplicity, we omitted the indices i and t from the variables.

Results indicate that the maximum projected payment of \$15,499.19 is expected to be collected for a property tax charge of \$722.06 per square meter.¹⁶ However, as the figure demonstrates, the local property tax between \$340 and \$1380 per square meter is off-sample, and for 95% of the full population, the property tax charge is below or equals to \$26.39 per square meter.¹⁷ The charges

¹⁷ The average data for these 83,511 panels are:

Panels	Years	Proj(PAYMENT)	PAYMENT	ARNONA/AREA
83,511	2013–2016	\$1264.77	\$1101.15	\$13.76

¹³ NIS is the local Israeli currency, where the conversion rate is 1 NIS roughly equals \$0.25. Referring to the units' area, the conversion rate for square meters is: 1 m² roughly equals 10.8 square feet.

¹⁴ ARNONA is the Hebrew terminology for local property tax charge. To reflect the taxpayer's perspective, we took the tax charge after discounts. Note, however, that the municipality has tighter control on the before discount tax charge and a limited control on discount rates, which are based on economic and socio-demographic criteria. Consequently, we ran robustness tests based on the before-discount tax charge.

¹⁵ Johnston and Dinardo, 1997 stress the importance of panel data estimation in econometrics. According to the authors: "Instead panel data estimation has grown in popularity because it has held out the promise of reducing a grave problem faced by most researchers: the lack of an adequate list of independent variables to explain the dependent variable" (page 395).

¹⁶ As a robustness test, we estimated the model based on the before-discount local property tax charge per square meter. Results indicate that The maximum projected payment of \$7641.98 is obtained for the property tax charge of $\frac{18.43}{2 \cdot 0.0128} = \720.92 per square meter.



Fig. 1. 2013–2016 Laffer curve of the local property tax in Jerusalem for all land uses. <u>Notes</u>: The upper figure displays the 2013–2016 Laffer Curve of the local property tax in Jerusalem for all land uses (with the exception of municipal assets) and for individuals who paid between 0% to 100% of the annual local property tax charge. The horizontal axis is the ARNONA (net local property tax charge) divided by the area, which is measured in square meters (1 m² = 10.7639 square feet). The vertical axis is the projected 2013–2016 annual payment. The lower figure displays the horizontal box plot of Arnona per square meter after omission of outliers. All charges and payments are originally measured in NIS (the local Israeli currency) and translated to US dollars (exchange rate of 1 NIS = 4 US Dollars). The figure is based on the following fixed effect regression outcomes for 87,711 panels (numbers in parentheses are standard errors, and all the coefficients are statistically significant at the 1% significance level): $proj(PAYMENT) = -2.84 \cdot 10^{-2} \left[\frac{ARNONA}{AREA} \right]^2 + 40.97 \left[\frac{ARNONA}{AREA} \right] + 707.73.$

 $(0.10 \cdot 10^{-2})$ (1.10) (18.06).

The maximum projected payment of \$15,499.19 is obtained for the property tax charge of $\frac{40.97}{2 + 0.0284} = 722.06 per square meter. The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(87710, 103752) = 23.93). This outcome supports the fixed-effect model. The Wu-Hausman test clearly rejects the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (chi2(2) = 908.58). The elasticity of the equation $Y = A_1 X^2 + B_1 X + C_1$ where Y = proj(PAYMENT) and $X = \left[\frac{ARNONA}{AREA}\right]$ is given by the formula: $E = \frac{2A_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$. The following table provides the elasticities for 95% of the full population of 196,915 observations (87,711 panels), who paid a local property tax below or equal to \$26.39 per square meter, and for the highest local property tax charge per square meter.

$X = \left[\frac{ARNONA}{AREA}\right]$	\$10.00	\$15.00	\$20.00	\$25.00	\$1,400
$E = \frac{2A_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$	0.3625	0.4573	0.5256	0.5768	-21.8890

of only one office exceeds \$1386 per square meter.¹⁸ These outcomes imply that, on the one hand, for most of the population, the policy of raising the local property tax each year is efficient. On the other hand, there is a clear tendency to avoid local property tax payments. For households, businesses or institutions whose after-discount charge is \$10, \$15, \$20 and \$25 per square meter, a 10% increase in the local property tax charge per square meter is expected to raise the projected payments by only 3.625%, 4.573%, 5.256% and 5.768%.¹⁹ Still, for a \$90 payment of a current tax charge of \$100, a 10% rise to \$110 is associated with a minimum and maximum anticipated rise to \$91.38 and \$92.76, respectively. In spite of the fact that anticipated collection rates dropped from 90% to 84.78–86.54%, the collected sum is nevertheless expected to rise.²⁰

Referring to the highest end of the most expensive local property tax per square meter, for the highest charge of \$1400 per square meter, a 10% *increase* is associated with 218.89% *drop* in the projected payment. Note, however, that the projected payment for the

¹⁸ T	'he	full	data	of	this	panel	are:
-----------------	-----	------	------	----	------	-------	------

Panels	Year	Proj(PAYMENT)	PAYMENT	ARNONA/AREA	AREA	ARNONA_NET
1	2015	\$1873.16	\$16,857.93	\$1415.09	12.0	\$16,981.08
1	2016	\$2961.67	\$17,024.80	\$1386.84	12.4	\$17,196.84

¹⁹ As a robustness test, we estimated the model based on the before-discount local property tax charge per square meter. Results indicate that for households, businesses or institutions, whose after-discount charge is \$10, \$15, \$20 and \$25 per square meter, a 10% increase in the local property tax charge per square meter is expected to raise the projected payments by only 1.538%, 2.127%, 2.63% and 3.064%.

tax charge per square meter is expected to raise the projected payments by only 1.538%, 2.127%, 2.63% and 3.064%. ²⁰ The minimum rise is: $\$00 \cdot \left[1 + \frac{3.625}{100}\right] = \93.2624 and the maximum rise is $\$90 \cdot \left[1 + \frac{5.78}{100}\right] = \95.1913 . The minimum and maximum collection rates are: $\frac{93.2624}{110} = \$4.78\%$ and $\frac{95.1913}{110} = \$6.54\%$, respectively. All the calculations are based on the assumption that the area of the asset remained unchanged during the sample period.



Fig. 2. Laffer curves of the local property tax for west vs. east Jerusalem. <u>Notes</u>: The upper (lower) figure refers to 76,713 (11,000) taxpayers in West (East) Jerusalem. The elasticities of payment with respect to local tax per square meter are given in the following table:

ARNONA	\$10.00	\$15.00	\$20.00	\$25.00	\$180
Elasticity West	0.3554	0.4498	0.5180	0.5695	0.7707
Elasticity East	0.5761	0.6480	0.6847	0.7027	-0.6988

\$1386.84–1415.09 charge for square meter equals \$1873.16–2961.67, which is much lower than the actual payment of \$16,857.93-\$17,024.80. This projection obeys the law of regression toward the mean (e.g., Liberman and Tversky, 1996).²¹ Yet, the projection provides a good approximation only for large number of panels. Consequently, a projection based on an outlier of one panel might not be reliable.²²

Fig. 2 stratifies the Laffer curve geographically between West and East Jerusalem. As noted in Section 3, the two major populations are Jews (61.4%), and Arab-Muslims (35.6%). These populations are economically segregated and also live in predominantly separate geographical locations, namely West and East Jerusalem (Alperovich & Deutsch, 1996).

Indeed, the outcomes reported in Fig. 2 reflect these differences in tax charges. For 75% of the full population of 170,407 (21,058) households, businesses or institutions in West (East) Jerusalem, the property tax charge is below or equals to \$21.65 (\$15.11) per square meter. Compared with West Jerusalem, the average property tax charge per square meter is lower significantly by \$6.08 at the 1% significance level. For the upper decile of property tax charge above \$ 80 per square meter, compared with West Jerusalem, the average property tax charge per square meter, compared with West Jerusalem, the average property tax charge above \$ 80 per square meter, compared with West Jerusalem, the average property tax charge and annual payment in West Jerusalem in 2015–2016 is \$1,386.84-\$1,415.09 per square meter, the equivalent

 $^{^{21}}$ According to the law, the projection is expected to become higher (lower) than the actual payment for values of the independent variable, which are above (below) the sample mean.

²² In this context, Liberman and Tversky (1996), and Kahneman (2002) pointed out that for a large number of observations sharing the same value of the independent variables, projected values (generated after removal of random components) provide better approximations of the dependent variable than the information they rely on. As Kahneman, 2002 notes: "As I understood clearly only when I taught statistics some years later, the idea that predictions should be less extreme than the information on which they are based is deeply counterintuitive."(Kahneman, 2002 – Biographical: The Military Experience, the end of the second paragraph, at:

http://www.nobelprize.org/nobel_prizes/economic-sciences/laureates/2002/kahneman-bio.html)

figures in East Jerusalem are only \$107.28-\$162.91.

Referring to the Laffer curve, while in West Jerusalem the local property tax tariff that yields the highest payment is \$723.13, in East Jerusalem the equivalent figure is \$140.89. The projected maximal payments are \$15,746.03 in West Jerusalem and only \$3,850.66 in East Jerusalem. For property tax charge of \$10, \$15, \$20 and \$25 the elasticity of average payment with respect to average after-discount net charge per square meter increases from 0.3554 to 0.5695 in West Jerusalem and from 0.5761 to 0.7027 in East Jerusalem. This implies an increasing tax collection rate with property tax charge per square meter in West (East) Jerusalem. Yet, for a 10% rise in local property tax charge per square meter, the maximum anticipated rise in payment is only 5.695% in West Jerusalem and 7.027% in East Jerusalem.

4. The Laffer curve: stratification based on land use

Table 2 displays the distribution of anticipated and actual tax collection, namely, the local property tax charge and the actual payment, stratified by land use and based on our panel dataset from 2013–2016. The budget is collected in NIS (the local Israeli currency). All the figures are converted to US dollars, where 1 NIS roughly equals \$0.25. The sum in US dollars for each land use is calculated as the product between the number of observations and the sample mean. Land uses are sorted in descending order based on the respective share of revenues from each of them.

Referring to the different land uses, the municipality produces the highest total revenues from the residential land use. According to our calculation in Tables 2, 78.47% of the revenues are produced from this use. This share is stratified to \$160,842,636.35 (\$45,453,285.80) originating from owner-occupiers (renters), and consisting of 61.18% (17.29%) of the total actual payments. Note, that unlike the United States, under the Israeli law, the local property tax charge is based on occupancy rather than ownership (e.g., Arbel et al., 2017). Consequently, in the normal state of events, municipal property tax charge is registered and paid directly by the Israeli renters and owner-occupiers.²³

Other land uses, which yield the highest revenues in descending order, include: office below and above 150 m^2 (7.92% and 7.86%); workshop and garage (1.25%); industry (1.20%); hotel (1.15%); public institution (0.82%); parking lot (0.63%); and government office (0.18%). The next step would thus be to construct estimated Laffer curves based on different land uses. Obviously, the projections obtained would become more reliable for large number of panels in sectors, such as, owner-occupiers (54,064 panels); renters (29,963 panels); office below and above 150 m^2 (2,838 and 410 panels, respectively); workshop and garage (317 panels); industry (123 panels); hotel (26 panels); and public institution (202 panels).

Table 2 displays the average collection rates for different land uses in 2013–2016. The $\left[\frac{PAYMENT}{ARNONA}\right]$ represents the current collection rate namely payment-property-tax-charge ratio, which is calculated for each observation separately. The $AVG\left[\frac{PAYMENT}{ARNONA}\right]$ is the average collection rate across all observations with the same land use. According to the table, collection rates are 37.72%-99.00%, where the lowest (highest) collection rate is obtained in.

Fig. 3 displays the estimated Laffer curve for owner-occupiers of dwelling units, who paid between 0% to 100% of the local property tax charge. Given our intent to avoid cases of arrears in debt payments, we excluded households who paid more than 100%. The upper figure is based on the fixed effect regression outcomes for 54,064 panels. Two tests support the fixed-effect model. The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(54,063, 73,355) = 33.84). Also, the results of the Wu-Hausman test clearly rejects the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (chi2(2) = 3893.88).

Results indicate that the maximum projected payment of \$3328.27 is obtained for the after-discount property tax charge of $\frac{76.14}{2 \cdot 0.48} = \79.98 per square meter, and the effective range of positive projections of payments is obtained for the after-discount local property tax charge of between \$0 and \$163.58 per square meter.²⁴ Based on the lower figure of the box-plot, excluding the small group of outliers, the on-sample panel is within this range of positive projections of payments, namely up to \$33. Finally, note that the corresponding elasticities for \$18.00, \$20.00, \$22.00 and \$24.00 after-discount tax charge per square meter are: 0.7078, 0.7064, 0.7023 and 0.6961. Elasticities, which are lower than 1 imply that for 10% rise in the charge per square meter the payment is expected to rise by less than 10%, namely by 7.078%, 7.064%, 7.023% and 6.961%, respectively.²⁵

Fig. 4 displays the estimated Laffer curve for renters of dwelling units, who paid between 0% to 100% of the local property tax charge. As noted, under Israeli law, local tax is paid on the basis of occupancy rather than ownership. Given our interest in filtering cases of arrears in debt payments, we excluded households who paid more than 100%. The upper figure is based on the fixed effect regression outcomes for 29,963 panels. Two tests support the fixed-effect model. The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(29,962, 25,625) = 6.99). Also, the Wu-Hausman test clearly rejects the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (chi2(2) = 571.04).

²³ American renters, for example, never see a tax bill and simply assume that they don't pay local property taxes – a phenomenon called "renter illusion" (see, for example, Banzhaf & Wallace, 2013; Brunner, Ross, Stephen, Becky & Simonsen, 2015).

²⁴ The maximum projected payment of \$3286.58 is obtained for the before-discount property tax charge of $\frac{102.87}{2 \cdot 0.72}$ = \$71.47 per square meter.

 $^{^{25}}$ Note that the corresponding elasticities for \$20.00 and \$24.00 before-discount tax charge per square meter are: 1.0736 and 0.9849, respectively. These higher estimates are plausible given that 1% rise in the after-discount tax charge is higher than 1% increase in the before-discount tax charge.



Fig. 3. Laffer curve of the local property tax in Jerusalem for owner-occupiers of dwelling units. <u>Notes</u>: The upper figure displays the 2013-2016 Laffer curve of the local property tax in Jerusalem for owner-occupiers who paid between 0% to 100% of the annual local property tax charge. The horizontal axis is the ARNONA (net local property tax charge) divided by the area, which is measured in square meters (1 square meter = 10.7639 square feet). The vertical axis is the projected 2013-2016 annual payment. The lower figure displays the horizontal box plot of Arnona per square meter after omission of outliers. All charges and payments are originally measured in NIS (the local Israeli currency) and translated to US dollars (exchange rate of 1 NIS = 4 US Dollars).

The figure is based on the following fixed effect regression outcomes for 54,064 panels (numbers in parentheses are standard errors, and all the coefficients are statistically significant at the 1% significance level): $proj(PAYMENT) = -0.48 \left[\frac{ARNONA}{AREA} \right]^2 + 76.14 \left[\frac{ARNONA}{AREA} \right] + 284.22.$ (0.01)(0.57)(6.45).

The maximum projected payment of \$3328.27 is obtained for the property tax charge of $\frac{76.14}{2 \cdot 0.48} = \79.98 per square meter. The effective range of positive projections is (-3.65, 0) and (163.58, 0). The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(54,063, 73,355) = 33.84). This outcome supports the fixed-effect model. The Wu-Hausman test clearly rejects the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (chi2(2) = 3893.88). The elasticity of the equation $Y = A_1X^2 + B_1X + C_1$ where Y = proj(PAYMENT) and $X = \left[\frac{ARNONA}{AREA}\right]$ is given by the formula: $E = \frac{2A_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$. The following table provides the elasticities for 95% of the full population of 127,421 observations (54,064 panels), who paid a local property tax below \$24 per square meter, and for the highest local property tax charge per square meter.

$X = \left[\frac{ARNONA}{AREA}\right]$	\$18.00	\$20.00	\$22.00	\$24.00	\$140
$E = \frac{2A_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$	0.7078	0.7064	0.7023	0.6961	-4.963

Results indicate that the maximum projected payment of \$1992.99 is obtained for the after-discount property tax charge of $\frac{49.73}{2 \cdot 0.35} = 70.35 per square meter, and the effective range of positive projections of payments is obtained for after-discount local property tax charge of between \$0 and \$145.44 per square meter.²⁶ Based on the lower figure of the box-plot, excluding the small group of outliers, the on-sample panel is within this range of positive projections of payments, namely below \$33. Finally, the corresponding elasticities for \$18.00, \$20.00, \$22.00 and \$24.00 after-discount tax charge per square meter are: 0.6504, 0.6491, 0.6446 and 0.6375.²⁷ Note, that compared with owner-occupiers, a 10% increase in the same after-discount local property tax charge for renters is expected to yield an average lower rise in payments of 6.454% (compared with 7.0315% for owner-occupiers). These outcomes are also in-line with the lowest collection rates from renters of 85.25% compared with 94.13% from owner-occupiers (see, for example, Table 2, and Arbel et al., 2017).

Figs. 5 and 6 display the estimated Laffer curve for individuals in the office sector, who paid between 0% to 100% of the local property tax charge. Fig. 4 refers to the small office sector below 150 m² and Fig. 5 covers the big office sector above 150 m². As discussed above, we excluded offices who paid more than 100%. The upper figures are based on the fixed effect regression outcomes for 2838 panels (small offices) and 410 panels (big offices). Two tests support the fixed-effect model. The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(2837, 3039) = 18.22 for small offices and F(409, 492) = 34.38 for big offices). Also, the Wu-Hausman test clearly rejects the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic only for small offices (chi2(2) = 5.17 for small offices) and chi2(1) = 2.22 and p-value = 13.65% for big offices).

Results indicate that for the sector of small offices the maximum projected payment of \$4068 is obtained for the after-discount property tax charge of $\frac{14.35}{2 \cdot 0.36} = \201.25 per square meter, and the effective range of positive projections of payments is obtained for after-discount local property tax charge of between \$0 and \$539.05 per square meter. As for the sector of big offices, the graph

²⁶ the maximum projected payment of \$1658.57 is obtained for the before-discount property tax charge of $\frac{42.84}{2 \cdot 0.31} =$ \$70.06 per square meter.

²⁷ Finally, the corresponding elasticities for \$20.00 and \$24.00 before-discount tax charge per square meter are: 0.6859 and 0.6691.



Fig. 4. Laffer Curve of the Local Property Tax in Jerusalem for Renters of Dwelling Units. <u>Notes</u>: The upper figure displays the 2013-2016 Laffer curve of the local property tax in Jerusalem for renters who paid between 0% to 100% of the annual local property tax charge. The horizontal axis is the ARNONA (net local property tax charge) divided by the area, which is measured in square meters (1 square meter = 10.7639 square feet). The vertical axis is the projected 2013-2016 annual payment. The lower figure displays the horizontal box plot of Arnona per square meter after omission of outliers. All charges and payments are originally measured in NIS (the local Israeli currency) and translated to US dollars (exchange rate of 1 NIS = 4 US Dollars).

The figure is based on the following fixed effect regression outcomes for 29,963 panels (numbers in parentheses are standard errors, and all the coefficients are statistically significant at the 1% significance level): $proj(PAYMENT) = -0.35 \left[\frac{ARNONA}{AREA}\right]^2 + 49.73 \left[\frac{ARNONA}{AREA}\right] + 243.55.$ (0.01)(1.07)(11.93).

The maximum projected payment of \$1992.99 is obtained for the property tax charge of $\frac{49.73}{2 \cdot 0.35} = 70.35 per square meter. The effective range of positive projections is (-4.74, 0) and (145.44, 0). The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(29,962, 25,625) = 6.99). This outcome supports the fixed-effect model. The Wu-Hausman test clearly rejects the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (chi2(2) = 571.04). The elasticity of the equation $Y = A_1X^2 + B_1X + C_1$ where Y = proj(PAYMENT) and $X = \begin{bmatrix} ARNONA \\ AREA \end{bmatrix}$ is given by the formula: $E = \frac{24_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$. The following table provides the elasticities for 92% of the full population of 55,590 observations (29,963 panels), who paid a local property tax below or equal to \$28 per square meter, and for the highest local property tax charge per square meter.

$X = \left[\frac{ARNONA}{AREA}\right]$	\$18.00	\$20.00	\$22.00	\$24.00	\$140
$E = \frac{2A_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$	0.6504	0.6491	0.6446	0.6375	-24.7650

increases monotonically, and the maximum projected payment is obtained for the outlier of \$180 after-discount net charge per square meter. Based on the lower figure of the box-plot, excluding the small group of outliers, the on-sample panel for both sectors is around after discount net tax tariffs of \$80-\$86 per square meter. Compared with the residential sector tariffs of \$18-\$24, these tariffs are very high. Finally, the corresponding elasticities for \$83, \$84, \$85 and \$86 after-discount tax charge per square meter are: 0.196, 0.1963, 0.1965 and 0.1966 for small office and .1.4756, 1.4848, 1.4938 and 1.5026 for big office land use categories.

The latter outcomes thus indicate that the efficiency associated with a rise in local property tax tariffs in the office sector is higher among big offices. Following a 10% increase of the after-discount net annual local property tax tariffs of \$83, \$84, \$85 and \$86 per square meter, payments of big offices with area above 150 m² are expected to rise by 14.756%, 14.848%, 14.938% and 15.026%,²⁸ and payments of small offices with area above 150 m² are expected to increase by only 1.960%, 1.963%, 1.965% and 1.966%.²⁹

5. Robustness test: first difference

One of the drawbacks associated with the analysis so far is the fact that the estimation procedure does not account for increases or reductions in local property tax rates across time. To remedy this drawback, we propose and apply the following fixed-effect model, which permits direct estimation of the payment-charge elasticity:

²⁸ Based on the collection rates from the office sector, for a \$85 payment of a current after-discount net tax charge of \$100, a 10% rise to \$110 is associated with a minimum and maximum anticipated rise to \$85. $\left[1+\frac{14.756}{100}\right]=$ \$97.5426 and \$85. $\left[1+\frac{15.026}{100}\right]=$ \$97.7721, respectively. These figures imply an increase in collection rates from 85% to $\frac{97.5426}{110} = 88.675\% - \frac{97.7721}{110} = 88.8837\%$

²⁹ Based on the collection rates from the office sector, for a \$85 payment of a current after-discount net tax charge of \$100, a 10% rise to \$110 is associated with a minimum and maximum anticipated rise to $\$5 \cdot \left[1 + \frac{1.960}{100}\right] = \86.666 and $\$5 \cdot \left[1 + \frac{1.960}{100}\right] = \86.6711 , respectively. These figures imply a drop in collection rates from 85% to $\frac{86.666}{110} = 78.787\% - \frac{86.6711}{110} = 78.792\%$

(3)

(4)



Fig. 5. Laffer curve of the local property tax in Jerusalem for small offices. <u>Notes</u>: The upper figure displays the 2013-2016 Laffer curve of the local property tax in Jerusalem for small offices below 150 square meters who paid between 0% to 100% of the annual local property tax charge and below \$350 per square meter. The horizontal axis is the ARNONA (net local property tax charge) divided by the area, which is measured in square meters (1 square meter = 10.7639 square feet). The vertical axis is the projected 2013-2016 annual payment. The lower figure displays the horizontal box plot of Arnona per square meter after omission of outliers. All charges and payments are originally measured in NIS (the local Israeli currency) and translated to US dollars (exchange rate of 1 NIS = 4 US Dollars).

The figure is based on the following fixed-effect regression outcomes for 2,838 panels (numbers in parentheses are standard errors, and all the coefficients are statistically significant at the 5%-1% significance levels): $proj(PAYMENT) = -3.57 \cdot 10^{-2} \left[\frac{ARNONA}{AREA} \right]^{2} + 14.35 \left[\frac{ARNONA}{AREA} \right] + 2,624.77.$ (1.43 · 10⁻²)(4.71)(294.54).

The maximum projected payment of \$4068.98 is obtained for the property tax charge of $\frac{14.35}{2\cdot0.36} = \201.25 per square meter. The effective range of positive projections is (-136.55, 0) and (539.05, 0). The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(2837, 3039) = 18.22). This outcome supports the fixed-effect model. The Wu-Hausman test clearly rejects the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (chi2(2) = 5.17). The elasticity of the equation $Y = A_1X^2 + B_1X + C_1$ where Y = proj(PAYMENT) and $X = \begin{bmatrix} \frac{ARNONA}{AREA} \end{bmatrix}$ is given by the formula: $E = \frac{24_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$. The following table provides the elasticities for 99.95% of the full population of 5879 observations (2838 panels), who paid a local property tax below or equal to \$400 per square meter, and for the highest local property tax charge per square meter.

$X = \left[\frac{ARNONA}{AREA}\right]$	\$83.00	\$84.00	\$85.00	\$86.00	\$390.00
$E = \frac{2A_1X^2 + B_1X}{A_1X^2 + B_1X + C_1}$	0.1960	0.1963	0.1965	0.1966	-1.8759

 $100 \cdot \Delta \ln (Y_{i,t}) = A_2 [100 \cdot \Delta \ln (X_{i,t})]^2 + B_2 [100 \cdot \Delta \ln (X_{i,t})] + C_2 + \mu_{2,i,t}$

$$\mu_{2,i,t} = \alpha_{2,i} + \epsilon_{2,i,t}$$

Where *i* is the index for each panel (individual or asset); *t* is the time index (t = 2013, 2014, 2015, 2016); $Y_{i,t} = (PAYMENT_{i,t})$, the annual payment; $\Delta \ln (Y_{i,t})$ is payment differences between two subsequent years for each panel, which provides an approximation to the variation in percent across years; $X_{i,t} = \begin{bmatrix} \frac{ARNONA_{i,t}}{AREA_{i,t}} \end{bmatrix}$, where $ARNONA_{i,t}$ is the net annual local property tax charge for asset *i* at year *t* converted to US dollars (exchange rate of 1 NIS = \$0.25), ³⁰ and $AREA_{i,t}$ is the area of asset *i* at year *t* measured in square meters ($1 \text{ m}^2 = 10.7639$ square feet); $\Delta \ln (X_{i,t})$ is the net annual tax charge differences between two subsequent years for each panel, which provides an approximation to the variation in percent across years; A_2 , B_2 , C_2 are parameters, $\alpha_{2,i}$ reflects a row vector of parameters (one parameter for each panel), and $\mu_{2,i,t}$ and $\varepsilon_{2,i,t}$ are the random disturbance terms. The parabolic model defined by Eq. (3) enables us to calculate the maximal point $(-\frac{B_2}{2A_2})$, and the elasticity of $Y_{i,t}$ with respect to $X_{i,t}$ $(\frac{\Delta \ln (Y_{i,t})}{\Delta \ln (X_{i,t})} = 2A_2 \Delta \ln(X_{i,t}) + B_2)$

Fig. 7 displays the estimation outcomes of Eqs. (3) and (4) for all sectors. The upper figure is based on the projected values for $-5 < 100 \cdot \Delta ln \left[\frac{ARNONA}{AREA}\right] < 10$ obtained from the fixed-effect regression outcomes for 69,472 panels whose collection rates are between 0% to 100%. The null hypothesis that the coefficients of all panels are equal is clearly rejected (F(69,471, 47,561) = 1.03 and the p-value is 0.00%). This outcome supports the fixed-effect model.

An interesting outcome obtained from the estimation of Eqs. (3) and (4) is the intersection points with the vertical and horizontal axes. On the one hand, the point (0, -3.3723) implies that if the net after-discount tax charge remains unchanged over time, projected payments are expected to drop by 3.3723% per annum. On the other hand, the point (4.1775, 0) indicates that for a 4.1775% net after-discount tax charge increase, projected payments remained unchanged. Consequently, the empirical model predicts that in order to maintain the same level of payments, an average 4.1775% increase is required per annum. Note also that payment-charge elasticity is around 80%, which is higher than the estimates obtained in previous section.

³⁰ Recall that ARNONA is the Hebrew terminology for local property tax charge.



Fig. 6. Laffer Curve of the Local Property Tax in Jerusalem for Big Offices. <u>Notes</u>: The upper figure displays the 2013-2016 Laffer curve of the local property tax in Jerusalem for big offices above 150 square meters who paid between 0% to 100% of the annual local property tax charge. The horizontal axis is the ARNONA (net local property tax charge) divided by the area, which is measured in square meters (1 square meter = 10.7639 square feet). The vertical axis is the projected 2013-2016 annual payment. The lower figure displays the horizontal box plot of Arnona per square meter after omission of outliers. All charges and payments are originally measured in NIS (the local Israeli currency) and translated to US dollars (exchange rate of 1 NIS = 4 US Dollars).

The figure is based on the following fixed effect regression outcomes for 410 panels (numbers in parentheses are standard errors, and the coefficient of the explanatory variable is statistically significant at the 1% significance level): $proj(PAYMENT) = 2.40 \left[\frac{ARNONA}{AREA} \right]^2 + 5,868.48$.

(0.68)(4,854.96).

The maximum projected payment of \$83,525.42 is obtained for the property tax charge of \$180 per square meter. The null hypothesis of equal coefficients for all panels is clearly rejected at the 1% significance level (F(409, 492)=34.38). This outcome supports the fixed-effect model. The Wu-Hausman test supports the null hypothesis that the differences between the coefficients obtained via the fixed-effect and random-effect procedures are not systematic (chi2(1) = 2.22 and p-value = 13.65%). The elasticity of the equation $Y = A_1X^2 + C_1$ where Y = proj(PAYMENT) and $X = \left[\frac{ARNONA}{AREA}\right]$ is given by the formula: $E = \frac{2A_1X^2}{A_1X^2 + C_1}$. The following table provides the elasticities for 99% of the full population of 903 observations (410 panels), who paid a local property tax below or equal to \$86 per square meter, and for the highest local property tax charge per square meter.

$X = \left[\frac{ARNONA}{AREA}\right]$	\$83.00	\$84.00	\$85.00	\$86.00	\$180
$E = \frac{2A_1 X^2}{A_1 X^2 + C_1}$	1.4756	1.4848	1.4938	1.5026	1.8595

The lower figure displays a box plot of the annual percentage of change in net local property tax charge per square meter during 2013–2016 and excluding outliers. The median increase is 1.26%, the increase in the third quartile is 3.30%, and the mean increase is 4.02%. The average increase is a bit lower than the projected increase of 4.1775% required to preserve the payments unchanged. Moreover, the distribution of the net local property tax charge is right-tailed, and the skewness is 2.26. This outcome may suggest that the municipality might encounter difficulties in preserving the current level of payments by raising each year the local property tax by 4.1775%.

Finally, note, that of the 76,698 panels in the sample, 10,546 panels (consisting of 13.75%) experience a decrease in local property tax after discounts, and only 2333 panels (consisting of 3.05%) experience a decrease in the local property tax before discounts. This suggests that for many households the source of this decrease in local property tax charges is additional discounts based on economics and socio-demographic criteria.

6. Summary and conclusion

The objective of the current study is to examine the efficiency of tax collection in the municipal level. This is done by constructing Laffer curves and estimating the elasticity of annual payment with respect to the after-discount net annual local property tax charge per square meter for different land uses and across time. No other paper we are aware of have employed this direct and tractable methodology. Given that practitioners are interested in tax collection rather than tax base, this methodology might prove to be useful. The relevant literature on Laffer curves have focused on income-tax brackets, labor and capital income tax, and the VAT in the US and European countries (e.g., Feldstein, 1995; Gruber and Saez, 2002; Trabandt and Uhlig, 2011; Oliviera, Francisca and Costa, 2015).

We employ a unique panel dataset referring to 2013–2016 obtained from Jerusalem municipality. The dataset contains microlevel information of land use, the annual local property tax charge and payment of each individual taxpayer, and the area of each asset measured in square meters. The local property tax can be examined in three dimensions: temporally, cross sectional based on different land uses, and geographically (West vs. East Jerusalem). During 2013–2016, the Jerusalem municipality steadily increased the local property tax rate by 0.46–1.42% per annum.³¹ Our study thus permits to examine the response of taxpayers to those

³¹ We also found a few cases of tax reductions.



Fig. 7. Elasticity of payments - local property tax charge 2013-2016. Notes: The upper figure displays the 2013-2016 relationship between the projected annual percentage of change in payment (vertical axis) and the annual percentage of change in local property tax charge per square meter (horizontal axis). The upper figure is based on the projected values for $-5 < 100 \cdot \Delta ln \left[\frac{ARNONA}{AREA}\right] < 10$ obtained from the following fixed-effect regression outcomes for 69.472 panels whose collection rates are between 0% to 100% (numbers in parentheses are standard errors and all the coefficients are statistically significant at the 1% significance level):.

$$proj[100 \Delta ln(PAYMENT)] = -3.787 \cdot 10^{-4} \left[100 \Delta ln \left[\frac{ARNONA}{AREA} \right] \right]^2 + 80.88 \cdot 10^{-2} \cdot 100 \Delta ln \left[\frac{ARNONA}{AREA} \right] - 337.23 \cdot 10^{-2} \cdot (0.88 \cdot 10^{-4}) (1.32 \cdot 10^{-2}) (1.32 \cdot 10^{-2}).$$

These outcomes imply that the intersection point with the horizontal axis is (4.1775, 0) based on the formula for the solution of the quadratic equation $\frac{-B - \sqrt{B^2 - 4AC}}{24}$ where = -3.787 $\cdot 10^{-4}$; $B = 80.88 \cdot 10^{-2}$ and $C = -337.23 \cdot 10^{-2}$, and the intersection point with the vertical axis is (0, -3.3723). The null hypothesis that the coefficients of all the panels are equal is clearly rejected (F(69,471, 47,561) = 1.03 and the p-value is 0.00%). This outcome supports the fixed-effect model.

The lower figure displays a box plot of the annual percentage of change in local property tax charge per square meter during 2013-2016 and excluding outliers. The median increase is 1.26%, the increase in the third quartile is 3.30% and the mean increase is 4.02%. Of the 76,698 panels in the sample, 10,546 panels (consisting of 13.75%) experience a decrease in local property tax after discounts, and only 2333 panels (consisting of 3.05%) experience a decrease in the local property tax before discounts.

Referring to Eq. (3), the payment-charge per square meters elasticity is given by the formula $\frac{\Delta \ln (Y)}{\Delta \ln (X)} = 2A_2 \Delta \ln(X) + B_2$ where Y = proj(PAYMENT)and $X = \left[\frac{ARNONA}{AREA}\right]$. The following table provides the on sample elasticities:

$100 \cdot \Delta \ln (X)$	-4	-2	0	2	4	6	8
$\frac{\Delta \ln (Y)}{\Delta \ln (X)} = 2A_2 \Delta \ln(X) + B_2$	0.8119	0.8103	0.8088	0.8073	0.8058	0.8043	0.8028

increases, and provides efficiency evaluation of this policy.

The outcomes imply that within the on-sample boundaries, raising the tax rate is efficient. For a \$90 payment of a current afterdiscount net tax charge of \$100, a 10% rise to \$110 is associated with a minimum and maximum anticipated rise to \$93.26 and \$95.19, respectively.³² In spite of the fact that anticipated collection rates dropped from 90% to 84.78%-86.54%, the overall collected sum is nevertheless expected to rise.33

Moreover, additional outcomes obtained from direct estimation of payment-property tax charge elasticities show that on the one hand, if local property tax charge remains unchanged, collection rates are anticipated to drop by 3.37% per annum. On the other hand, if the municipality would like to preserve the same collected sum of payments as before, for the entire sample, the minimal anticipated required raise is 4.17% per annum. Finally, within the on-sample range (between -5% and 10% variation in the net local property tax), the elasticity of payment-property tax charge is around 0.8. This implies that the projected increase in payment following a 10% increase in after-discount net local property tax charge is around 8%.

Additional findings, which refer separately to the two sectors yielding 94% of the total revenues, namely, the residential and office sectors, suggest that increasing local property tax tariffs are more efficient for owner-occupiers rather than for renters, and for big offices rather than small ones. Research findings may thus be of assistance to municipal policy planners.

³² As demonstrated in Table 2, the initial tax collection rate for all sectors is 91.22% ³³ The minimum rise is: \$90: $\left[1+\frac{3.625}{100}\right] = 93.2624 and the maximum rise is \$90: $\left[1+\frac{5.768}{100}\right] = 95.1913 . The minimum and maximum collection rates are: $\frac{93.2624}{110} = 84.78\%$ and $\frac{95.1913}{110} = 86.54\%$, respectively. All the calculations are based on the assumption that the area of the asset remained un-changed during the sample period.

Appendix A. Israel: Regional division to statistical areas

See Appendix Fig. A1.



Fig. A1. Source: Beenstock, Felsenstein and Zeev (2011); Fig. 1, page 604.

Appendix B. Local authorities: Socio economic clusters of population

See Appendix Fig. B1.

אפיון יחידות גיאוגרפיות וסיווגן לפי הרמה החברתית-כלכלית של האוכלוסייה 2008 - ריי



Fig. B1. Source: Israeli Central Bureau of Statistics: Characterization and Classification of Geographical Units by the Socio-Economic Level of the Population 2008, at: http://www.cbs.gov.il/hodaot2013n/24_13_087map.pdf. The map describes the local authorities socio-economic clusters of population in Israel from the lowest cluster 1–2 () to the highest cluster 9–10 ().

Number of Residents	80% discount for income	60% discount for income		40% discount for income		
	Up to	From	Up to	From	Up to	
1	2,703 NIS	2,704 NIS	3,108 NIS	3,109 NIS	3,513 NIS	
2	4,054 NIS	4,055 NIS	4,662 NIS	4,663 NIS	5,270 NIS	
3	4,703 NIS	4,704 NIS	5,408 NIS	5,409 NIS	6,113 NIS	
4	5,351 NIS	5,352 NIS	6,154 NIS	6,155 NIS	6,957 NIS	
5	6,535 NIS	6,536 NIS	7,515 NIS	7,516 NIS	8,496 NIS	
6	7,719 NIS	7,720 NIS	8,877 NIS	8,878 NIS	10,035 NIS	
Number of Residents	80% discount for income	80% discount fo	r income	80% discount fo	or income	
_	Up to	From	Up to	From	Up to	
7	8,903 NIS	8,904 NIS	10,238 NIS	10,239 NIS	11,574 NIS	
8	10,087 NIS	10,088 NIS	11,600 NIS	11,601 NIS	13,113 NIS	
9	11,271 NIS	11,272 NIS	12,961 NIS	12,962 NIS	14,652 NIS	
10 and up	1,252 NIS per person	1,440 NIS per pe	erson	1,628 NIS per p	erson	

Appendix C. Percentage of discount on the local property tax residential tariffs: low income households

Notes: Source of Information: Jerusalem Municipality website at:

https://www.jerusalem.muni.il/en/Municipality/Services/Arnona/Discounts/Income/Pages/default.aspx

The table displays municipal tax discounts for income reasons for the fiscal year 2015. Discounts are determined based on the number of persons living in the dwelling unit, and the average monthly income during the fourth quarter in 2014. NIS is the local Israeli currency, where 1 NIS roughly equals \$0.25.

Appendix D. Local property tax residential tariffs in Jerusalem per square meter (NIS)

D1. 2017 Tariffs

		Zone			
Building type	Criteria	Α	В	С	D
1 2 3 4	Apartment of more than 120 m ² Apartment of up to 120 m ² Apartment with external/joints lavatories Apartment in building made of wood or tin	107.69 NIS 89.23 NIS 62.76 NIS 38.44 NIS	97.86 NIS 71.57 NIS 46.67 NIS 38.44 NIS	75.50 NIS 53.26 NIS 38.44 NIS 38.44 NIS	61.71 NIS 38.44 NIS 38.44 NIS 38.44 NIS

D2. 2013 Tariffs

		Zone			
Building type	Criteria	А	В	С	D
1 2 3 4	Apartment of more than 120 m ² Apartment of up to 120 m ² Apartment with external/joints lavatories Apartment in building made of wood or tin	100.36 NIS 83.15 NIS 58.49 NIS 35.84 NIS	91.20 NIS 66.70 NIS 43.50 NIS 35.84 NIS	70.36 NIS 49.64 NIS 35.84 NIS 35.84 NIS	57.51 NIS 35.84 NIS 35.84 NIS 35.84 NIS

Notes: Source of this information: Jerusalem Municipality website:

https://www.jerusalem.muni.il/en/municipality/services/arnona/rates/pages/tariffs%20 for%20 residential%20 assets.aspx and

https://www.jerusalem.muni.il/Residents/MunicipalityServices/PropertyTax/taxorder/Documents/tax2013.pdf

NIS is the local Israeli Currency, where 1 NIS roughly equals US \$ 0.25. 1 sq meter roughly equals 10.7639 square feet. The table implies a 7.25%-7.31% increase in local tax charge per square meter during a period of five-years. This implies a $[(1 + 0.0725)^{0.2} - 1] = 1.41\%^{-1}$ a $[(1 + 0.0731)^{0.2} - 1] = 1.42\%$ increase per annum

Appendix E. Tax rate tarrifs for non-residential assets in Jerusalem per 1 sqm

Type of use	Zone	2013 Tariff Per 1	2017 sam	% change
	20110		- -	// enange
Services and commercial offices (above 150 sqm)	All city	326.07 NIS	333.72 NIS	2.34%
Services and commercial offices (up to 150 sqm)	All city	315.95 NIS	323.35 NIS	2.34%
Call Centers (not banks)	All city		135.51 NIS	
Workshops and garages	Area A	143.94 NIS	154.46 NIS	7.30%
	Area B	106.92 NIS	114.74 NIS	7.30%
Industrial alarta	Area C	86.76 NIS	93.09 NIS	7.30%
	Area A	120.28 INIS	135.51 NIS	7.31%
	Area B	93.70 NIS	100.54 NIS	7.30%
Schools kindergartens, vechivas and universities	All city	70.00 MIS	01.01 NIS	7.30%
Banke	All city	1264 21 NIS	1256 70 NIS	7.3270
Main/regional bank management	All city	886 59 NIS	951 51 NIS	7.32%
Bank services call center		422 16 NIS	453 07 NIS	7.32%
Central banking services branches	All city	422.10 MI3	707 40 NIS	7.3270
Stock exchange	All city	1233 87 NIS	1324 23 NIS	7 32%
Occupied land		37 93 NIS	40 69 NIS	7.32%
occupica iana	Area B	33 08 NIS	35 48 NIS	7.20%
	Area C	21 74 NIS	23 22 NIS	7.20%
Clinics health funds and public hospitals	All city	07 58 NIS	104 72 NIS	7 3 20%
Places of worship	All city	58 54 NIS	62 82 NIS	7 31%
Insurance companies	All city	443 51 NIS	475 98 NIS	7 32%
Consulates embassies and UN institutions	All city	97 58 NIS	104 72 NIS	7.32%
Hotels and boarding houses with an area of more than 2000 sam	Area A	132 80 NIS	142 51 NIS	7 31%
notes and boarding nouses with an area of more than 2000 squi	Area B	123 67 NIS	132 70 NIS	7 30%
	Area C	99 70 NIS	106 99 NIS	7.30%
Hotels and hoarding houses with an area of up to 2000 sam	Area A	107 81 NIS	115 69 NIS	7 31%
notes and boarding nouses with an area of up to 2000 squi	Area B	87 33 NIS	93 70 NIS	7 29%
	Area C	63 49 NIS	68 13 NIS	7.20%
Old age homes	Area A	100 36 NIS	107 69 NIS	7 30%
old age nomes	Area B	91 20 NIS	97 86 NIS	7.30%
	Area C	70.36 NIS	75 50 NIS	7.30%
	Area D	57 51 NIS	61 71 NIS	7 30%
Museums	All city	97 58 NIS	104 72 NIS	7 32%
Bowling gyms and sports halls	All city	166 85 NIS	179.05 NIS	7 31%
Sports center with an area over 10.000 m^2 Tariff for 1000 first square meters.	All city	166 85 NIS	179.05 NIS	7 31%
Sports center with an area over $10,000 \text{ m}^2$ from 1001 sam to 2000 sam	All city	100.00 110	89 52 NIS	7.0170
Sports center with an area over 10,000 m ² from 2001 sqm to 2000 sqm	All city		66 62 NIS	
Large Laundromats services with an area of more than 400 som	i ili city		00.02 110	
in the Atarot Industrial area (parcel 510 and 710)	All city	108 62 NIS	116 56 NIS	7 30%
Marketing warehouses located at the Atarot Industrial area	All city	108.62 NIS	116.56 NIS	7.30%
Cinemas and theaters up to 300 som	All city	110.45 NIS	118.53 NIS	7.32%
Cinemas and theaters from 301 som to 1000 som	All city	73.63 NIS	79.01 NIS	7.31%
Cinemas and theaters with more than 1001 som	All city	36.81 NIS	39.48 NIS	7.25%
Electricity facility	All city	162.51 NIS	174.41 NIS	7.32%
Structure serving as a closed water reservoir used by the Gibon Spring	All city	62.61 NIS	67.18 NIS	7.30%
Paid Parking Lots	Area A	59.41 NIS	63.74 NIS	7.29%
0	Area B	39.74 NIS	42.64 NIS	7.30%
	Area C	30.90 NIS	33.14 NIS	7.25%
Free Parking space open to the public	Area A	25.57 NIS	27.43 NIS	7.27%
	Area B	20.46 NIS	21.88 NIS	6.94%
	Area C	15.35 NIS	16.46 NIS	7.23%
Indoor Parking space which is not open to the public	Area A	25.57 NIS	27.43 NIS	7.28%
	Area B	20.46 NIS	21.88 NIS	6.94%
	Area C	15.35 NIS	16.46 NIS	7.23%
Studio artists and sculptors to 100 sqm	All citv	54.67 NIS	58.67 NIS	7.32%
	,			

Studio Painters and Sculptors began sqm 101onwards	All city 163.96 NIS	175.97 NIS 7.32%
Farmland	All city 0.59 NIS	0.63 NIS 6.78%
Agricultural Structure	All city 38.87 NIS	41.70 NIS 7.28%
Banquet halls	All city	179.05 NIS
Other assets	All city 326.07 NIS	339.62 NIS 4.16%

Source: Tarrifs for non-Residential Assets, Jerusalem Municipality Website at:

https://www.jerusalem.muni.il/en/municipality/services/arnona/rates/pages/non-residentialassets.aspx

https://www.jerusalem.muni.il/Residents/MunicipalityServices/PropertyTax/taxorder/Documents/tax2013.pdf

Notes: NIS is the local Israeli Currency, where 1 NIS roughly equals US \$ 0.25. 1 sq meter roughly equals 10.7639 square feet

References

Alperovich, Gershon, & Deutsch, Joseph (1996). Urban structure with two coexisting and almost completely segregated populations: The case of East and West Jerusalem. Regional Science and Urban Economics, 26(2), 171–187.

Andreoni, J., Erard, B., & Feinstein, J. (1998). Tax compliance. Journal of Economic Literature, 36(2), 818-860.

Arbel, Yuval, Fialkoff, Chaim, & Kerner, Amichai (2017). Removal of renter's illusion: Property tax compliance among renters and owner-occupiers. Regional Science and Urban Economics, 66, 150–174.

Banzhaf, H. Spencer, & Wallace, E. Oates (2013). On fiscal illusion in local public finance: Re-examining Ricardian equivalence and the renter effect. National Tax Journal, 66(3), 511–540.

Beenstock, Michael, Felsenstein, Daniel, & Zeev, Nadav Ben (2011). Capital deepening and regional inequality: An empirical analysis. *The Annals of Regional Science*, 47(3), 599–617.

Brunner, Eric J., Ross, Stephen L., & Simonsen, Becky K. (2015). Homeowners, renters and the political economy of property taxation. Regional Science and Urban Economics, 53, 38–49.

Oliviera, De, Francisca, Guedes, & Costa, Leonardo (2015). The VAT Laffer curve and the business cycle in the EU27: An empirical approach. *Economic Issues*, 20(2), 29–44.

Feldstein, Martin (1995). The effect of marginal tax rates on taxable income: A panel study of the 1986 tax reform act. *Journal of Political Economy*, *103*(3), 551–572. Fischel, William (1992). Property taxation and the tiebout model: Evidence for the benefit view from zoning and voting. *Journal of Economic Literature*, *30*(1), 171–177. Fullerton, Don (1982). On the possibility of inverse relationship between tax rates and government revenues. *Journal of Public Economics*, *19*, 3–22.

Fullerton, Don (2008). Laffer curve, The new Palgrave Dictionary of Economics. In Steven N. Durlauf, & Lawrence E. Blume (Eds.). The New Palgrave Dictionary of Economics onlinePalgrave Dictionary Macmillanhttps://doi.org/10.1057/9780230226203.0922.

Greene, William H. (2012). Econometric analysis (7th ed.). Boston and London: Pearson Education Limited. http://www.worldcat.org/title/econometric-analysis/oclc/ 726074601.

Gruber, Jon, & Saez, Emanuel (2002). The elasticity of taxable income: Evidence and implications. Journal of Public Economics, 84, 1-32.

Haughwout, Andrew, Inman, Robert, Craig, Steven, & Luce, Thomas (2004). Local revenue hills: Evidence from four U.S. cities. The Review of Economics and Statistics, 86(2), 570–585.

Johnston, Jack, & Dinardo, John (1997). Econometric Methods (4th ed.). McGraw Hill International Edition.

Kahneman, D. (2002) Biographical: The military experience, at: http://www.nobelprize.org/nobel_prizes/economic-sciences/laureates/2002/kahneman-bio.html. Kastlunger, Barbara, Kirchler, Erich, Mittone, Luigi, & Pitters, Julia (2009). Sequence of audits, tax compliance, and taxpaying strategies. *Journal of Economic Psychology*, 30, 405–418.

Kirchler (2007). The economic psychology of tax behaviour. Cambridge: Cambridge University Press.

Laffer, A. (2004). The Laffer Curve: Past, Present, and Future, Backgrounder, # 1765: 1–16. Available at: https://web.archive.org/web/20071212035655/http://www.heritage.org/Research/Taxes/upload/64214_1.pdf.

Liberman, Varda, & Tversky, Amos (1996). Critical thinking: Statistical reasoning and intuitive judgment (Hebrew). Israel: The Open University of Israel.

Mayo, Stephen, K. (1981). Theory and esatimation in the economics of housing demand. Journal of Urban Economics, 10, 95-116.

Rees-Jones, Alex (2018). Quantifying Loss-Averse Tax Manipulation. *The Review of Economic Studies*, 85(2), 1251–1278. https://doi.org/10.1093/restud/rdx038. Stuart, C. E. (1981). Swedish tax rates, labor supply and tax revenues. *Journal of Political Economy*, 89, 1020–1038.

Saez, E., Slemrod, J., & Giertz, S. H. (2012). The elasticity of taxable income with respect to marginal tax rates: A critical review. Journal of Economic Literature, 50(1), 3–50. https://doi.org/10.1257/jel.50.1.3.

Trabandt, Mathias, & Uhlig, Harald (2011). The Laffer curve revisited. Journal of Monetary Economics, 58, 305-327.

Wanniski, Jude (1978). Taxes, Revenues, and the 'Laffer Curve. *The Public Interest*, 3–16 Available at: https://www.nationalaffairs.com/storage/app/uploads/public/58e/1a4/c54/58e1a4c549207669125935.pdf).

Yitzhaki, Shlomo, Golan, Yolanda, & Aviad Tur-Sinai (2013). Municipal authorities ranking - by the socio-economic index. Working Paper Series, number 77: 1–40 (Hebrew). Available at: http://www.publications/pw77.pdf>.